Author's response to reviews

Title: Piper sarmentosum inhibits ICAM-1 and Nox4 gene expression in oxidative stress-induced human umbilical vein endothelial cells

Authors:

Azizah Ugusman (dr.azizah@yahoo.com)
Zaiton Zakaria (zaitonukm@gmail.com)
Chua Kien Hui (ckienhui@hotmail.com)
Nor Anita Megat Mohd Nordin (noranitamn@gmail.com)

Version: 3 Date: 2 April 2011

Author's response to reviews: see over
Dear BMC Series Editorial Production Team,

Thank you for your valuable comments. I have complied with all the major and minor revisions needed for the publication. The followings are the corrections that have been made in the manuscript.

Major revision:

I have indicated under authors' contributions that ‘all authors read and approved the final manuscript’.

Minor revisions:

- Capital letters are used only for the first word and proper nouns in the title (*Piper sarmentosum* inhibits ICAM-1 and Nox4 gene expression in oxidative stress-induced human umbilical vein endothelial cells).

- The text 'Original Research Article' has been removed from the title page.

- All text is displayed in black throughout the manuscript.

- Authors' contributions are formatted into a paragraph.

- The visible vertical lines have been removed from the table.

- All figures are consistently mentioned as Fig.

- Typography errors have been carefully checked and there is no typography error in the manuscript.

I am sure you will accept all the changes made in the revised version. Once again, I wish to thank you for the valuable comments.

Best regards,

Dr. Zaiton Zakaria